\documentclass[reqno,12pt,a4]{amsart}
\NeedsTeXFormat{LaTeX2e}[1994/12/01]
%\usepackage{vmargin}
%\setpapersize{A4}
\textheight23cm
\textwidth17cm
\addtolength{\topmargin}{-50pt}
\addtolength{\oddsidemargin}{-1.8cm}
\addtolength{\evensidemargin}{-1.8cm}

\usepackage{amssymb}
\usepackage{amsmath}

\usepackage[all]{xy}
%\CompileMatrices

\def\Hom{\mathop{\rm Hom}\nolimits}
\def\Ext{\mathop{\rm Ext}\nolimits}
\def\id{\mathop{\rm id}\nolimits}
\def\End{\mathop{\rm End}\nolimits}
\def\Ker{\mathop{\rm Ker}\nolimits}
\def\Im{\mathop{\rm Im}\nolimits}
\def\Ad{\mathop{\rm Ad}\nolimits}
\def\tr{\mathop{\rm tr}\nolimits}
\def\ev{\mathop{\rm ev}\nolimits}
\def\diag{\mathop{\rm diag}\nolimits}
\def\dist{\mathop{\rm dist}\nolimits}
\def\supp{\mathop{\rm supp}\nolimits}
\def\sgn{\mathop{\rm sgn}\nolimits}
\def\e{{\varepsilon}}
\def\t{{\tau}}
\def\th{{\theta}}
\def\b{{\beta}}
\def\la{{\langle}}
\def\ra{{\rangle}}
%\def\v{{\vert}}
\def\V{{\Vert}}
%\def\ar{{\ \longrightarrow\ }}
\def\a{{\alpha}}
\def\d{{\delta}}
\def\g{{\gamma}}
\def\l{{\lambda}}
\def\ph{{\varphi}}
\def\s{{\sigma}}
\def\i{{\infty}}
\def\D{{\Delta}}
\def\G{{\Gamma}}
\def\q{{\hfill$\square$}}

%\def\q{{\hfill $\Box$}}%\par\medskip}}
\def\proof{{\bf Proof. }}


\parindent=0em
\setlength{\parskip}{\medskipamount}


\newtheorem{thm}{Theorem}[section]
\newtheorem{lem}[thm]{Lemma}
\newtheorem{prop}[thm]{Proposition}
\newtheorem{dfn}[thm]{Definition}
\newtheorem{cor}[thm]{Corollary}
\newtheorem{conj}[thm]{Conjecture}
\newtheorem{rmk}[thm]{Remark}
\newtheorem{prob}[thm]{Problem}
\newtheorem{obs}[thm]{Observation}

%\renewcommand{\theequation}{\thesection.\arabic{equation}}
\numberwithin{equation}{section}

%\newcommand{\norm}[1]{\left\| #1 \right\|}
\newcommand{\norm}[1]{\| #1 \|}
\newcommand{\ov}[1]{\overline{#1}}
\newcommand{\til}[1]{\widetilde{#1}}
\newcommand{\cone}{\operatorname{cone}}
\newcommand{\co}{\operatorname{co}}
\newcommand{\im}{\operatorname{im}}
\newcommand{\Bott}{\operatorname{Bott}}
\newcommand{\Aut}{\operatorname{Aut}}
\newcommand{\as}{\operatorname{as}}
\renewcommand{\theenumi}{\roman{enumi}}
\renewcommand{\labelenumi}{$\theenumi)$}
%\renewcommand{\theenumii}{\roman{enumii}}
\renewcommand{\labelenumii}{$\theenumii)$}

\def\A{{\mathcal A}}
\def\H{{\mathcal H}}
\def\K{{\mathcal K}}
\def\B{{\mathcal B}}
\def\M{{\mathcal M}}
\def\T{{\mathcal T}}
\def\S{{\mathcal S}}
\def\C{{$C^*$-algebra}}

%\usepackage{showkeys}
\sloppy

%\date{\today}
\author{V. Manuilov and K. Thomsen}
\title[Translation invariant asymptotic homomorphisms]{Extensions of $C^*$-algebras and translation invariant asymptotic homomorphisms}



\begin{document}
\begin{thebibliography}{99}


\bibitem{A} W. Arveson, {\it Notes on extensions of $C\sp*$-algebras}, Duke Math. J. {\bf 44} (1977), 329--355.

\bibitem{An}
{J. Anderson},
{\it A $C^*$-algebra for which $\Ext(A)$ is not a group},
Annals of Math. {\bf 107} (1978), 455--458.

%\bibitem{BDF1}
%{L.G. Brown, R.G. Douglas and P.A. Fillmore},
%{\it Unitary equivalence modulo the compact operators and
%extensions of $C^*$-algebras}, Proc. Conf. on Operator Theory,
%Lecture Notes in Mathematics {\bf 345}, Springer Verlag (1973), 58--128.



\bibitem{BDF2}
{L.G. Brown, R.G. Douglas and P.A. Fillmore},
{\it Extensions of $C^*$-algebras and $K$-theory}, Ann. Math.
{\bf 105} (1977), 265--324.


\bibitem{CH}
{A. Connes and N. Higson},
{\it D\'eformations, morphismes asymptotiques et
$K$-th\'eorie bivariante}, C. R. Acad. Sci. Paris S\'er. I Math. {\bf 311}
(1990), 101--106.

%\bibitem{CuHi}
%{J. Cuntz, N. Higson.}
%{\it Kuiper's theorem for Hilbert modules},
%Operator algebras and mathematical physics.
%Contemp. Math. {\bf 62}, Amer. Math. Soc., 1987, 429--435.




%\bibitem{Cu} {J. Cuntz}, {\it A new look at KK-theory}, K-theory {\bf 1} (1987), 31-51.

%\bibitem{CS} J. Cuntz and G. Skandalis, {\it Mapping cones and exact sequences in KK-theory}, J. Operator Theory {\bf 15} (1986), 163-180.



%\bibitem{DL} {M. D\u ad\u arlat and T.A. Loring}, {\it $K$-homology, asymptotic representations and unsuspended $E$-theory}, J. Funct. Anal. {\bf 126} (1994), 367-383.


%\bibitem{HT} {U. Haagerup and S. Thorbj\o rnsen},
%{\it A new application of random matrices:
%$\Ext\left( C^*_{red}(F_2)\right)$ is not a group},
%Preprint, SDU Odense, 2002.



\bibitem{H} N. Higson, {\it On the $K$-theory proof of the index theorem},  Index theory and operator algebras (Boulder, CO, 1991),  67--86, Contemp. Math., 148, Amer. Math. Soc., Providence, RI, 1993. 
 
\bibitem{HKT} N. Higson, G. Kasparov and J. Trout, {\it A Bott periodicity theorem for infinite-dimensional Euclidian space}, Adv. Math. {135}, (1998), 1-40.


%\bibitem{H1} N. Higson, {\it A characterization of KK-theory}, Pacific J. Math. {\bf 126} (1987), 253-276.

%\bibitem{H2} \bysame , {\it Algebraic $K$-theory of stable
%$C\sp *$-algebras}, Adv. Math. {\bf 67} (1988), no. 1, 140 pp.


\bibitem{H-LT} T. Houghton-Larsen, K. Thomsen, {\em  Universal (co)homology theories}, K-theory {\bf 16} (1999), 1-27.



%\bibitem{Kapl}
%{A. Kaplan}, {\it Remark on certain $C^*$-algebra extensions considered by
%G. Skandalis}, Proc. Amer. Math. Soc. {\bf 117} (1993), 1047--1050.

\bibitem{K}{G.G. Kasparov}, {\it The operator $K$-functor and extensions of $C^*$-algebras},  Izv. Akad. Nauk SSSR Ser. Mat. {\bf 44}(1980), 571--636.


\bibitem{Ke}{J. L. Kelley, I. Namioka}, {\it Linear Topological Spaces}, van Nostarnd, New York, 1963. 


\bibitem{Kirch} {E. Kirchberg}, {\it On non-semisplit extensions,
tensor products and exactness of group $C^*$-algebras},
Invent. Math. {\bf 112} (1993), 449-489.

%\bibitem{K-JT} {K. Knudsen-Jensen and K. Thomsen},
%{\it Elements of $KK$-theory}, Birkh\"auser, Boston, 1991.


%\bibitem{Lance}
%{E. C. Lance}, {\it Hilbert $C^*$-modules. A toolkit for operator
%algebraists}. London Math. Soc. Lecture Note Series {\bf 210}. Cambridge
%University Press, Cambridge, 1995.

%\bibitem{L}
%{T. Loring}, {\it Almost multiplicative maps between
%$C^*$-algebras}, Operator Algebras and Quantum Field Theory.
%Internat. Press, 1997, 111--122.


%\bibitem{MT1}
%{V. M. Manuilov, K. Thomsen}, {\it Quasidiagonal extensions and
%sequentially trivial asymptotic homomorphisms}. Adv. Math. {\bf 154}
%(2000), 258--279.



\bibitem{MT3}
{V. M. Manuilov, K. Thomsen}, {\it The Connes--Higson construction is an isomorphism}. Preprint.


\bibitem{MT5}
{\bysame}, {\it $E$-theory is a special case of
$KK$-theory}. Proc. London Math. Soc., to appear.

\bibitem{MT4}
{\bysame}, {\it Semi-invertible extensions and asymptotic homomorphisms}. Preprint.


%\bibitem{M}
%{V. M. Manuilov}, {\it Asymptotic homomorphisms into the Calkin algebra}.
%J. Reine Angew. Math., to appear.

\bibitem{R} W. Rudin, {\it Functional Analysis}, McGraw-Hill, New Delhi, 1973.


%\bibitem{S1}  {G. Skandalis}, {\it Une notion de nucl\'earit\'e en K-th\'erie (d'apr\`es J. Cuntz)}, K-theory {\bf 1} (1988), 549-573.

%\bibitem{S2} \bysame, {\it Le bifuncteur de Kasparov
%n'est pas exact}, C.R. Acad. Sci. Paris, S\'er. I Math.
%{\bf 313} (1991), 939--941.


%\bibitem{T}
%{K. Thomsen}, {\it Homotopy classes of $\ast$-homomorphisms between
%stable $C^{\ast}$-algebras}. Duke Math. J. {\bf 61} (1990), 67--104.

%\bibitem{T}
%{K. Thomsen}, {\it Homotopy invariance in E-theory}.
%Preprint. \AA rhus, 2001.

\bibitem{Th?} K. Thomsen, {\it Equivariant $KK$-theory and $C^*$-extensions}, K-theory {\bf 19} (2000), 219-249.


%\bibitem{V} D. Voiculescu, {\it A note on quasidiagonal $C^*$-algebras and homotopy}, Duke Math. J.  {\bf 62} (1991), 267-271.



%\bibitem{Wass}
%{S. Wassermann}, {\it $C^*$-algebras associated with  groups with
%Kazhdan's property $T$}. Ann. Math. {\bf 134} (1991), 423--431.

\end{thebibliography}

\maketitle

\begin{abstract}
Let $A$, $B$ be $C^*$-algebras; $A$ separable, $B$ $\sigma$-unital and stable.
We introduce a notion of translation invariance for asymptotic homomorphisms from $SA=C_0(\mathbb R)\otimes A$ to $B$ and show that the Connes--Higson construction applied to any extension of $A$ by $B$ is homotopic to a translation invariant asymptotic homomorphism. In the other direction we give a construction which produces extensions of $A$ by $B$ out of such a translation invariant asymptotic homomorphism. This leads to our main result; that the homotopy classes of extensions coincide with the homotopy classes of translation invariant asymptotic homomorphisms.

\end{abstract}

\section{Introduction}

The excision properties of the E-theory of Connes and Higson hinges on a fundamental construction, introduced in \cite{CH}, which associates an asymptotic homomorphism $SA \to B$ to an extension of the separable $C^*$-algebra $A$ by another $\sigma$-unital $C^*$-algebra $B$. This construction transforms an extension into something which seems to be of a different nature, in much the same way as the Busby-invariant transforms extensions, viewed as short exact sequences of $C^*$-algebras, into $*$-homomorphisms. Both constructions have turned out to be very useful for the study of extensions of $C^*$-algebras, and in this way also for many other purposes. But while it is easy to see that the Busby-invariant holds all relevant information on the extension, it is far more difficult to decide how much information is lost when considering the asymptotic homomorphism arising from an extension, rather than the extension itself. This question is quite intriguing because the approach to the study of extensions which is based solely on the Busby-invariant, i.e. the approach developed by Brown, Douglas and Fillmore in \cite{BDF2}, and later in higher generality by Kasparov \cite{K}, is now known to have serious deficiencies: It may not have a group structure, \cite{An}, not even in cases where the homotopy classes of extensions \emph{do} have such a structure, \cite{Kirch}. In our previous work, \cite{MT3,MT4}, we have been able, in certain cases, to determine exactly which structure the asymptotic homomorphism arising from the Connes-Higson construction encodes. But the general case has escaped us, and the results of the present paper show why. As we shall explain the asymptotic homomorphism $SA \to B$ has a property which has been overlooked so far; it is namely \emph{translation invariant} in the sense that a shift in the parameter of the asymptotic homomorphism corresponds exactly to a shift of the same amount in $SA$. This observation shows that the parameter of the asymptotic homomorphism is not just a structureless dummy, whose only raison d'\^etre is to approach infinity without actually getting there, but that it has a serious relationship to the $C^*$-algebra $SA$ on which the asymptotic homomorphism is defined, at least when it arises from the Connes-Higson construction. Besides making this observation precise, we will show that this additional structure of the asymptotic homomorphism is exactly what is needed in order to recover the extension, showing that there are no more surprises to be uncovered.    



Let $A$ and $B$ be $C^*$-algebras. We say that a one-parameter family of maps, $\psi = (\psi_t)_{t \in \mathbb R} : A \to B$, is an \emph{asymptotic homomorphism} when $(\psi_t)_{t \in [1,\infty)}$ is (in the usual sense, \cite{CH}) and $\lim_{t\to-\infty}\psi(a)=0$ for all $a\in A$. $\psi$ is \emph{uniformly continuous} when the function $t \mapsto \psi_t(a)$ is uniformly continuous for all $a \in A$, and \emph{equi-continuous} when the family of maps $\psi_t : A \to B, t \in \mathbb R$, is. 

Let $\tau=(\tau_s)_{s\in\mathbb R}$ be the action of $\mathbb R$ on $SA = C_0(\mathbb R, A)$ by translations,
$$
\tau_s(f)(t) = f(t -s),\quad f\in SA,
$$ 
and consider an asymptotic homomorphism $\psi = (\psi_t)_{t \in \mathbb R} : SA \to B$. We say that $\psi$ is \emph{asymptotically translation invariant} when $\lim_{t \to \infty} \psi_{t-s}(f) - \psi_t \circ \tau_s(f) =0$ for all $s \in \mathbb R$ and all $f \in SA$, and \emph{translation invariant} when $\psi$ is equicontinuous, and
$$
\psi_t \circ \tau_s = \psi_{t-s}
$$
for all $t,s \in \mathbb R$. Two (asymptotically) translation invariant asymptotic homomorphisms, $\varphi$ and $\psi$, are \emph{homotopic} when there is a (asymptotically) translation invariant asymptotic homomorphism $\Phi : SA \to IB (=C[0,1]\otimes B)$ such that $\ev_1 \circ \Phi_t = \varphi_t$ and $\ev_0 \circ \Phi_t = \psi_t$ for all $t \in \mathbb R$, where $\ev_s : IB \to B$ denotes evaluation at $s \in [0,1]$. We denote by $[[SA,B]]_{\tau}$ the set of homotopy classes of translation invariant asymptotic homomorphisms $SA \to B$, and by $[[SA,B]]_{a,\tau}$ the set of homotopy classes of asymptotically translation invariant asymptotic homomorphisms $SA \to B$. There is then a natural map
$$
[[SA,B]]_{\tau} \to [[SA,B]]_{a,\tau}.
$$
It will be a corollary of our main result that this map is an isomorphism of abelian semigroups when $A$ is separable and $B$ $\sigma$-unital and stable.






\section{From extensions to translation invariant asymptotic homomorphisms}\label{vch}


Let $A$ and $B$ be $C^*$-algebras; $A$ separable, $B$ $\sigma$-unital. The multiplier algebra of $B$ will be denoted by $M(B)$ and the generalized Calkin algebra $M(B)/B$ by $Q(B)$. Let $\psi : A \to Q(B)$ be an extension. We choose
\begin{enumerate}
\item[a)] a strictly positive element $b_0 \in B$ such that $0 \leq b_0 \leq 1$,
\item[b)] a continuous and homogeneous lift $s : A \to M(B)$ of $\psi$, and
\item[c)] a sequence $h_0 \leq h_1 \leq h_2 \leq \dots$ of functions in $C_0(0,1]$,
\end{enumerate}
such that 
\begin{enumerate}
\item[d)] $h_{n+1}h_n = h_n$ for all $n$,
\end{enumerate}
and the sequence $u_n = h_n(b_0), n = 0,1,2, \dots$, has the properties
\begin{enumerate}
\item[i)] $ \lim_{n \to \infty} u_ns(a) - s(a)u_n = 0$ for all $a \in A$, and
\item[ii)] $\lim_{n \to \infty} u_nb =b$ for all $b \in B$.
\end{enumerate}
This is possible by the arguments that prove the existence of quasi-central approximate units, \cite{A}. $\{u_n\}_{n=0}^{\infty}$ is a \emph{unit sequence} in the sense of \cite{MT3} and \cite{MT5}.



We set $\Delta_0 = \sqrt{u_0}$ and $\Delta_n = \sqrt{u_n - u_{n-1}}, n \geq 1$. A crucial observation is that thanks to i) we have that
\begin{equation}\label{e13}
\Delta_i\Delta_j =0, \ \text{when $|i-j| \geq 2$.}
\end{equation}
Let $t_0 \leq t_1 \leq t_2 \leq \dots$ be a sequence in $[0, \infty[$ such that
\begin{enumerate}
\item[iii)] $\lim_{n \to \infty} t_n = \infty$,
\item[iv)] $\lim_{n \to \infty} t_n - t_{n+1} = 0$.
\end{enumerate}

The following lemma appears in \cite{MT5}.

\begin{lem}\label{mt5} For any norm-bounded sequence $\{m_j\} \subseteq M(B)$, and for any $k\in\mathbb N$, the sum $\sum_{j=0}^{\infty} \Delta_j m_j
\Delta_{j+k}$ converges in the strict topology to an element of $M(B)$, and
$$
\left\|\sum_{j=0}^{\infty} \Delta_j m_j \Delta_{j+k} \right\| \leq \sup_j \|m_j\|.
$$
Furthermore, $\sum_{j=0}^{\infty} \Delta_j m_j \Delta_{j+k} \in B$ when $\lim_{j \to \infty} \|m_j\| = 0$.
\end{lem}




For each $t \in \mathbb R$, define $\varphi_t : SA \to B$ by
\begin{equation}\label{ugh}
\varphi_t(f)= \sum_{j=0}^{\infty} \Delta_j s\left(f(t -t_j)\right)\Delta_j .
\end{equation}
The sum converges in norm because $\lim_{j \to \infty} s\left(f(t -t_j)\right) = 0$, cf. Lemma \ref{mt5}.

Since $s$ is homogeneous and continuous at $0$, there is an $L > 0$ such that $\|s(a)\| \leq L\|a\|$ for all $a \in A$. It follows that
\begin{equation}
\left\|\varphi_t(f)\right\| \leq L\|f\|
\end{equation}
for all $f$ and $t$, cf. Lemma \ref{mt5}.

\begin{lem}\label{a1}

$\left(\varphi_t\right)_{t \in \mathbb R}$ is a translation invariant asymptotic homomorphism. I.e.

\begin{enumerate}
\item[A)] $\left(\varphi_t\right)_{t \in [0,\infty)}$ is an asymptotic homomorphism,
\item[B)] $\varphi_t \circ \tau_s = \varphi_{t-s}$ for all $t,s \in \mathbb R$,
\item[C)] $\lim_{t \to -\infty}\varphi_t(f) = 0$ for all $f \in SA$,
\item[D)] $\varphi$ is equicontinuous.
\end{enumerate}


\end{lem}
\begin{proof} B) is a trivial observation and C) follows from Lemma \ref{mt5} since $\lim_{x \to -\infty} s(f(x)) = 0$, so it remains to prove A) and D). We begin with A): Let $f,g\in SA$ and let $\epsilon > 0$. By uniform continuity of the map $x \mapsto s(g(x))$, there is a $\delta > 0$ so small that $\left\|s(g(x)) - s(g(y))\right\| < \epsilon$ when $|x-y| < \delta$. Choose $N \in \mathbb N$ so large that $t_j - t_{j-1} < \delta$ when $j \geq N$. Since $\lim_{t \to \infty} s(k(t-t_j)) = 0$ for all $j \in \mathbb N$ and all $k \in SA$, there is then a $T \in \mathbb N$ so large that
$$
\left\|\varphi_t(f) -  \sum_{j=N}^{\infty} \Delta_j s\left(f(t -t_j)\right)\Delta_j \right\| < \epsilon,
\qquad
\left\|\varphi_t(g) -  \sum_{j=N}^{\infty} \Delta_j s\left(g(t -t_j)\right)\Delta_j \right\| < \epsilon,
$$
and
$$
\left\|\varphi_t(fg) -  \sum_{j=N}^{\infty} \Delta_j s\left(fg(t -t_j)\right)\Delta_j \right\| < \epsilon
$$
when $t \geq T$.
In what follows we use the notation $a\sim_\epsilon b$ when $\|a-b\|\leq\epsilon$.
Then
\begin{align*}
\varphi_t(f)\varphi_t(g)
&\sim_{(\|f\| + \|g\|)L\epsilon} \\ \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)\right)\Delta_j^2s\left(g(t - t_j)\right)&\Delta_j +   \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)\right)\Delta_j\Delta_{j+1}s\left(g(t - t_{j+1})\right)\Delta_{j+1}   \\ & + \sum_{j=N+1}^{\infty} \Delta_j s\left(f(t-t_j)\right)\Delta_j\Delta_{j-1}s\left(g(t - t_{j-1})\right)\Delta_{j-1}\\
\end{align*}
when $t \geq T$. It follows from i) that if $N$ is large enough, then
$$
\sup_{x \in \mathbb R} \left\|\left[\Delta_j^2,s\left(g(x))\right)\right]\right\| < \epsilon ,
\qquad
\sup_{x \in \mathbb R} \left\|\left[\Delta_j\Delta_{j\pm 1},s\left(g(x)\right)\right]\right\| < \epsilon,
$$
when $j \geq N$. So by increasing $N$ if necessary, we may assume, thanks to Lemma \ref{mt5}, that
\begin{align*}
 \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)\right)\Delta_j^2s\left(g(t - t_j)\right)\Delta_j +   \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)\right)\Delta_j\Delta_{j+1}s\left(g(t - t_{j+1})\right)\Delta_{j+1}   \\ + \sum_{j=N+1}^{\infty} \Delta_j s\left(f(t-t_j)\right)\Delta_j\Delta_{j-1}s\left(g(t - t_{j-1})\right)\Delta_{j-1}\\  \\
\sim_{\epsilon} \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)\right)s\left(g(t - t_j)\right)\Delta_j^3 +   \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)\right)s\left(g(t - t_{j+1})\right)\Delta_j\Delta_{j+1}^2   \\ + \sum_{j=N+1}^{\infty} \Delta_j s\left(f(t-t_j)\right)s\left(g(t - t_{j-1})\right)\Delta_j\Delta_{j-1}^2\\  \\
\end{align*}
when $t \geq T$. It follows from ii) that
$$
\lim_{j \to \infty} \sup_{x \in \mathbb R} \left\|\Delta_j^2 \left[ s(f(x))s(g(x)) - s(f(x)g(x))\right]\right\| = 0
$$
so by increasing $N$ further, and using Lemma \ref{mt5} again, we can arrange that
\begin{align*}
 \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)\right)s\left(g(t - t_j)\right)\Delta_j^3 +   \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)\right)s\left(g(t - t_{j+1})\right)\Delta_j\Delta_{j+1}^2   \\ + \sum_{j=N+1}^{\infty} \Delta_j s\left(f(t-t_j)\right)s\left(g(t - t_{j-1})\right)\Delta_j\Delta_{j-1}^2   \\ \sim_{\epsilon}
\sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_j)\right)\Delta_j^3 +   \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_{j+1})\right)\Delta_j\Delta_{j+1}^2   \\ + \sum_{j=N+1}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_{j-1})\right)\Delta_j\Delta_{j-1}^2  .
\end{align*}
when $t \geq T$. By using iv) we can also arrange that
\begin{align*}
\sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_j)\right)\Delta_j^3 +   \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_{j+1})\right)\Delta_j\Delta_{j+1}^2   
\\
 + \sum_{j=N+1}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_{j-1})\right)\Delta_j\Delta_{j-1}^2  
\\
\sim_{\epsilon}\sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_j)\right)\Delta_j^3 +   \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_{j})\right)\Delta_j\Delta_{j+1}^2   \\ + \sum_{j=N+1}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_{j})\right)\Delta_j\Delta_{j-1}^2 
\end{align*} 
when $t \geq T$. Since $\Delta_j^3 + \Delta_j\Delta_{j+1}^2 + \Delta_j\Delta_{j-1}^2 = \Delta_j$, we can finally arrange that
\begin{align*}
\sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_j)\right)\Delta_j^3 +   \sum_{j=N}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_{j})\right)\Delta_j\Delta_{j+1}^2  
\\
 + \sum_{j=N+1}^{\infty} \Delta_j s\left(f(t-t_j)g(t - t_{j})\right) 
\Delta_j\Delta_{j-1}^2  
\\
\sim_{\epsilon}  \sum_{j=N}^{\infty} \Delta_j s(fg(t -t_j))\Delta_j 
\end{align*} 
when $t \geq T$. All in all it follows that there is a $T \in \mathbb R$ so that
$$
 \varphi_t(f)\varphi_t(g) \sim_{(\|f\| + \|g\|)L\epsilon + 5\epsilon} \varphi_t(fg)
$$
for all $t \geq T$. The asymptotic linearity and self-adjointness of $\varphi$ follows in the same way.

D): Let $\epsilon > 0$. For each $x \in A, \delta > 0$, let $B(x,\delta) = \{a \in A: \ \|a-x\| < \delta\}$. Since $K = f(\mathbb R)$ is a compact subset of $A$, and $s$ is continuous, there are finite sets, $\{x_1,x_2, \dots, x_N\} \subseteq K$ and $\{\delta_1,\delta_2, \dots, \delta_N\} \subseteq ]0,1[$, such that $K \subseteq \bigcup_{i=1}^N B(x_i, \delta_i)$ and $a,b \in B(x_i,2 \delta_i) \Rightarrow \|s(a) - s(b)\| < \epsilon $ for all $i$. Set $\delta = \min \{\delta_1,\delta_2, \dots, \delta_N\}$. It follows then from Lemma \ref{mt5} that when $g \in SA$ and $\|f-g\| < \delta$, then $\left\|\varphi_t(f)-\varphi_t(g)\right\| \leq \epsilon$ for all $t \in \mathbb R$.




\end{proof}


The next aim is to show that up to homotopy of translation invariant asymptotic homomorphisms, $\varphi$ depends only on $\psi$.



\subsection{Independence of the choice of $\{t_n\}_{n=0}^{\infty}$.}



When $\{t_n'\}_{n=0}^{\infty}$ is another sequence in $[0,\infty[$ such that iii) and iv) hold, we define $\Phi_t^{\lambda} : SA \to B$ by
$$
\Phi_t^{\lambda}(f) = \sum_{j=0}^{\infty} \Delta_j s(f(t-\lambda t_j - (1-\lambda)t_j'))\Delta_j .
$$

To see that $[0,1] \ni \lambda \mapsto \Phi_t^{\lambda}(f)$ is continuous, let $\epsilon > 0$ be given. Since $\lim_{j \to \infty} t_j = \lim_{j \to \infty} t_j' = \infty$, there is an $N$ such that $\left\|\Phi_t^{\lambda}(f) - \sum_{j=0}^{N} \Delta_j s(f(t-\lambda t_j - (1-\lambda)t_j'))\Delta_j\right\| < \epsilon$ for all $\lambda \in [0,1]$. Since  $[0,1] \ni \lambda \mapsto  \sum_{j=0}^{N} \Delta_j s(f(t-\lambda t_j - (1-\lambda)t_j'))\Delta_j$ is clearly continuous, we see that $[0,1] \ni \lambda \mapsto \Phi_t^{\lambda}(f)$ is continuous.


We can then define $\Phi_t : SA \to IB (= C[0,1]\otimes B)$ such that
$$
\Phi_t(f)(\lambda) = \Phi_t^{\lambda}(f) =  \sum_{j=0}^{\infty} \Delta_j s(f(t-\lambda t_j - (1-\lambda)t_j'))\Delta_j .
$$
The arguments that proved Lemma \ref{a1} show that $\Phi = \left(\Phi_t\right)_{t \in \mathbb R}$ is a translation invariant asymptotic homomorphism. It clearly defines a homotopy of translation invariant asymptotic homomorphisms connecting the translation invariant asymptotic homomorphisms arising from the two choices of sequences $\{t_n\}_{n=0}^{\infty}$ and $\{t_n'\}_{n=0}^{\infty}$.




\subsection{Independence of the choice of unit sequence.}



Let $g_0 \leq g_1 \leq g_2 \leq \dots$ be another sequence of functions in $C_0(0,1]$ such that d) holds, and $v_n = g_n(b_0), n = 0,1,\dots $, satisfies i)-ii). We can then find a third sequence $k_0 \leq k_1 \leq k_2 \leq \dots$ in $C_0(0,1]$ such that d) holds and the sequence $w_n = k_n(b_0), n = 0,1,2, \dots$, also satisfies i)-ii), and at the same time that
$$
w_nu_n = u_n, \ w_nv_n = v_n
$$
for all $n$. Set $d_n = 1 - \sum_{j=n+1}^{\infty} 2^{-j}, n = 0,1,2, \dots$, and define $W_n \in C[0,1]\otimes B$ such that
$$
W_n(\lambda) = \begin{cases} u_n, & \lambda \in [0,d_n],  \\ \frac{d_{n+1} - \lambda}{d_{n+1} -d_n}u_n + \frac{\lambda - d_n}{d_{n+1} -d_n}w_n, & \lambda \in [d_n,d_{n+1}], \\ w_n, & \lambda \in [d_{n+1},1] . \end{cases}
$$
Define $\Phi_t : SA \to IB$ such that
$$
\Phi_t(f) = \sqrt{W_0}s(f(t-t_0))\sqrt{W_0} + \sum_{j=1}^{\infty} \sqrt{W_j - W_{j-1}}s(f(t-t_j)) \sqrt{W_j - W_{j-1}}.
$$
The arguments of Lemma \ref{a1} show that $\Phi = \left(\Phi_t\right)_{t \in \mathbb R}$ is a translation invariant asymptotic homomorphism, so it gives us a homotopy of translation invariant asymptotic homomorphisms connecting the translation invariant asymptotic homomorphisms arising from the two choices of sequences $\{u_n\}_{n=0}^{\infty}$ and $\{w_n\}_{n=0}^{\infty}$. Using $\{v_n\}_{n=0}^{\infty}$ instead of $\{u_n\}_{n =0}^{\infty}$ in the construction of $\Phi$, we get a homotopy of translation invariant asymptotic homomorphisms connecting the translation invariant asymptotic homomorphisms arising from the two choices of sequences $\{v_n\}_{n=0}^{\infty}$ and $\{w_n\}_{n=0}^{\infty}$. By transitivity of homotopy, we see that the two translation invariant asymptotic homomorphisms arising from $\{v_n\}_{n=0}^{\infty}$ and $\{u_n\}_{n=0}^{\infty}$ are homotopic.







\subsection{Independence of the choice of strictly positive element $b_0$.}\label{subsec}


Let $b_0'$ be another strictly positive element of $B$. Then
$$
c_0(\lambda) = \lambda b_0 + (1 -\lambda)b_0'
$$
is a strictly positive element of $IB$. Constructing an asymptotic homomorphism $SA \to IB$ from $c_0$ as in Lemma \ref{a1}, we obtain a homotopy of translation invariant asymptotic homomorphisms connecting two translation invariant asymptotic homomorphisms $SA \to B$; one arising from the construction by use of $b_0$, the other by use of $b_0'$.

\subsection{Independence of the choice of section $s$.}

When $s' : A \to M(B)$ is another continuous and homogeneous lift of $\psi$ we can choose the sequence $h_0 \leq h_1 \leq h_2 \leq \dots$ such that
$$
\lim_{j \to \infty} \sup_{x \in \mathbb R}\left\|\Delta_j\left(s(f(x)) - s'(f(x))\right)\right\| = 0
$$
for all $f \in SA$. It follows then (from Lemma \ref{mt5}) that
$$
\lim_{t \to \infty} \left[\sum_{j=0}^{\infty} \Delta_j s(f(t-t_j))\Delta_j - \sum_{j=0}^{\infty} \Delta_j s'(f(t-t_j))\Delta_j \right] = 0 
$$
for all $f \in SA$.


\bigskip

\subsection{Homotopy invariance.}

Let $\psi':A\to Q(B)$ be an extension weakly homotopic to $\psi$. Then there exist an extension $\Psi:A\to Q(IB)$ with evaluation maps at the endpoints coinciding with $\psi$ and $\psi'$, respectively. Once more, constructing an asymptotic homomorphism $SA \to IB$ from $\Psi$ as in Lemma \ref{a1} gives us a homotopy of translation invariant asymptotic homomorphisms.


We can now summarize in the following:


\begin{thm}\label{a4} The homotopy class of the translation invariant asymptotic homomorphism $\varphi$ of Lemma \ref{a1} depends only on the weak homotopy class of $\psi$ in the set of extensions of $A$ by $B$.
\end{thm}

In other words, if we denote by $\Ext_h(A,B)$ the (weak) homotopy classes of extensions of $A$ by $B$, the formula (\ref{ugh}) gives rise to a map
$$
CH_{\tau} : \Ext_h(A,B) \to [[SA,B]]_{\tau}.
$$







\section{Relation to the Connes-Higson construction.}

In this section we show that up to homotopy $CH_{\tau}(\psi)$ agrees with the Connes-Higson construction $CH(\psi)$, as it was introduced in \cite{CH}.

Let
$$
\cone = \{f \in C_b(\mathbb R): \ \lim_{t \to - \infty} f(t) = 0, \ \text{and} \ \lim_{t \to \infty} f(t) \ \text{exists} \ \},
$$
and
$$
I = \{f \in C_b(\mathbb R): \ \lim_{t \to \pm \infty} f(t)  \ \text{exist} \ \ \}.
$$
Then the sum
$$
\sum_{j=0}^{\infty} \Delta_j^2 h(t-t_j)
$$
converges in the strict topology to an element $\lambda_t(h) \in M(B)$ for all $t \in \mathbb R$ and every $h \in I$, cf. Lemma \ref{mt5}.

Note that $\lambda_t : I \to M(B)$ is a linear completely positive contraction for all $t$. We claim that
\begin{equation}\label{20}
\lim_{t \to \infty} \lambda_t(f)\lambda_t(g) - \lambda_t(fg) = 0
\end{equation}
for all $f,g \in I$. To see this, observe that
\begin{eqnarray*}
&& \Bigl(\sum_{j=0}^{\infty} \Delta_j^2 f(t-t_j) \Bigr) \Bigl(\sum_{j=0}^{\infty} \Delta_j^2g(t-t_j)\Bigr) \\
%&\\
&& = \sum_{j=0}^{\infty} \Delta_j^4 f(t-t_j )g(t-t_j) +  \sum_{j=0}^{\infty} \Delta_j^2\Delta_{j+1}^2f(t-t_j )g(t-t_{j+1}) \\
&& \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ +
\sum_{j=1}^{\infty} \Delta_{j}^2\Delta_{j-1}^2 f(t-t_{j} )g(t-t_{j-1})
\ \ \ \ \ \ \ \ \ \ \ \ \ \ \text{(by (\ref{17}))} \\
\end{eqnarray*}
If $t$ is big enough,
$$
\sup_{j \geq 0} |f(t-t_{j} )g(t-t_{j+1}) -  f(t-t_{j} )g(t-t_j)| \leq \epsilon
$$
and
$$
\sup_{j\geq 1} |f(t-t_{j} )g(t-t_{j-1}) -  f(t-t_{j} )g(t-t_j)| \leq \epsilon .
$$
Hence Lemma \ref{mt5} shows that
\begin{eqnarray*}
&& \\ \lambda_t(f)\lambda_t(g)
%&\\
&& \sim_{2\epsilon} \sum_{j=0}^{\infty} \Delta_j^4 f(t-t_j)g(t-t_j) +  \sum_{j=0}^{\infty} \Delta_j^2\Delta_{j+1}^2f(t-t_j)g(t-t_{j}) \\
&& \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ +
\sum_{j=1}^{\infty} \Delta_{j}^2\Delta_{j-1}^2 f(t-t_{j} )g(t-t_{j}) \\
&&
= \lambda_t(fg)
\end{eqnarray*}
for all large enough $t$, i.e. (\ref{20}) holds. Note that $\lambda_t \circ \tau_s = \lambda_{t-s}$. Furthermore, it is obvious that $\lim_{t \to -\infty} \lambda_t(f) = 0$ for all $f \in S = C_0(\mathbb R)$, and that $\lim_{t \to \pm \infty} \lambda_t(g)b = g(\pm \infty)b$ for all $b \in B$ and all $g \in I$. In fact, $\lim_{t \to -\infty} \lambda_t(h) = h(-\infty)$, in norm. Furthermore, $\lambda_t(f) - f(-\infty)1 \in B$ for all $t,f$. To sum up:




\begin{lem}\label{a7} There is a completely positive and translation invariant asymptotic homomorphism $\lambda = \left(\lambda_t\right)_{t \in \mathbb R}: I \to M(B)$ such that
\begin{enumerate}
\item[1)] $\lambda_t(h) =\sum_{j=0}^{\infty} \Delta_j^2 h(t-t_j), \ t \in \mathbb R, h \in I$,
\item[2)] $\lambda_t(h) - h(-\infty)1 \in B , \ t \in \mathbb R, h \in I$,
\item[3)] $\lambda_t \circ \tau_s = \lambda_{t-s}, \ t,s \in \mathbb R$,
\item[4)] $\lim_{t \to -\infty} \lambda_t(h) = h(-\infty)1, \ h \in I$,
\item[5)] $\lim_{t \to \infty}\lambda_t(h)b = h(\infty)b, \ h \in I, b \in B$.
\end{enumerate}
\end{lem}




Since $A$ is separable there is a compact subset $X \subseteq A$ with dense span in $A$.


\begin{lem}\label{a5}


The sequence $h_0 \leq h_1 \leq h_2 \leq \cdots $ can be chosen so that
\begin{equation}\label{e6}
\sum_{n=0}^{\infty} \left\|\left[\Delta_n,s(a)\right]\right\| < \infty
\end{equation}
for all $a \in X$, and
\begin{equation}\label{e7}
s(a) - \sum_{j=0}^{\infty} \Delta_j s(a)\Delta_j \in B
\end{equation}
for all $a \in A$.
\end{lem}
\begin{proof} As is well-known there is a sequence $\delta_1 > \delta_2 > \delta_3 > \dots $in $]0,1[$ such that
$$
b \in B , \ 0 \leq b \leq 1, \  \|b s(a) - s(a)b\| < \delta_n \ \forall a \in X \  \Rightarrow \ \|\sqrt{b} s(a) - s(a)\sqrt{b}\| < 2^{-n} \ \forall a \in X .
$$
If we therefore choose $h_n$, as we can, such that $ \|u_n s(a) - s(a)u_n\| < {\delta_n}$ for all $n \in \mathbb N$ and all $a \in X$, we find that (\ref{e6}) holds. Note that
$$
s(a) -  \sum_{j=0}^{\infty} \Delta_j s(a)\Delta_j = \sum_{j=0}^{\infty} \Delta_j^2s(a) -  \Delta_j s(a)\Delta_j = \sum_{j=0}^{\infty} \Delta_j \left[\Delta_j, s(a)\right],
$$
for all $a \in A$, with strictly convergent sums. It follows from (\ref{e6}) that the last sum is convergent in norm when $a \in X$, so that $s(a) - \sum_{j=0}^{\infty} \Delta_j s(a)\Delta_j \in B$ for all $a \in X$. Since $s$ is linear modulo $B$ and since $X$ spans a dense subspace of $A$, we conclude that $s(a) - \sum_{j=0}^{\infty} \Delta_j s(a)\Delta_j \in B$ for all $a \in A$.
\end{proof}














Let $\kappa : \mathbb R \to ]0,1[$ be a continuous increasing function such that $\lim_{t \to - \infty} \kappa(t) = 0$ and $\lim_{t \to \infty} \kappa(t) = 1$. Set
$$
v_t = \sum_{j=0}^{\infty} \Delta_j^2 \kappa(t - t_j).
$$


\begin{lem}\label{a3} Assume that $h_0 \leq h_1 \leq h_2 \leq \dots$ is chosen so that (\ref{e6}) and (\ref{e7}) hold. Then

\begin{enumerate}
\item[a)] $\lim_{t \to -\infty}v_t = 0$,
\item[b)] $\lim_{t \to \infty} v_tb = b$ for all $b \in B$,
\item[c)] $\lim_{t \to \infty} v_ts(a) -s(a)v_t = 0$ for all $a\in A$,
\item[d)] $\lim_{t\to \infty}  g(v_t)s(a) - \varphi_t((g \circ \kappa) \otimes a) = 0$ for all $g \in C_0]0,1[, a \in A$.
\end{enumerate}
\end{lem}
\begin{proof} a) follows from Lemma \ref{mt5} because $\lim_{x \to -\infty} \kappa(x) = 0$. b) : Let $b \in B$ and $\epsilon > 0$ be given. There is an $N$ so large that $u_Nb \sim_{\epsilon} b$. Since
$$
\lim_{t \to \infty} v_tu_Nb = \lim_{t \to \infty} \sum_{j=0}^{N+1}\Delta_j^2\kappa(t-t_j)u_Nb =\sum_{j=0}^{N+1}\Delta_j^2u_Nb = u_Nb,
$$
we conclude that $v_tb \sim_{3 \epsilon} b$ for all $t$ large enough.

c): Since $s(a) - \sum_{j=0}^{\infty}\Delta_js(a)\Delta_j \in B$ for all $a \in A$, it suffices, thanks to b), to show that
\begin{equation}\label{e1}
\lim_{t \to \infty} v_t \left(\sum_{j=0}^{\infty}\Delta_js(a)\Delta_j\right) - \left(\sum_{j=0}^{\infty}\Delta_js(a)\Delta_j\right)v_t = 0.
\end{equation}
Let $\epsilon > 0$. Choose a $\delta > 0$ so small that $x,y \in \mathbb R, |x-y| < \delta \Rightarrow |\kappa(x) -\kappa(y)| < \epsilon$, and let $K \in \mathbb N$ be so large that $t_{n+2} - t_n < \delta$ when $n \geq K-1$. Set $\tilde{\kappa} = 1 - \kappa$ and note that $1 - v_t = \sum_{j=0}^{\infty} \Delta_j \tilde{\kappa}\left(t-t_j\right)\Delta_j$. Since $\lim_{x \to \infty} \tilde{\kappa}(x) = 0$ there is a $T \in \mathbb R$ so large that $\left\|1- v_t - \sum_{j=K}^{\infty} \Delta_j^2\tilde{\kappa}(t-t_j)\right\| < \epsilon$ when $t \geq T$. It follows that
\begin{equation*}
\begin{split}
&\left( 1 -v_t\right)\left(\sum_{j=0}^{\infty}\Delta_js(a)\Delta_j\right) \sim_{L \|a\| \epsilon} \sum_{j=K}^{\infty} \left( \Delta_j^2 \tilde{\kappa}(t - t_j) + \Delta_{j-1}^2\tilde{\kappa}(t-t_{j-1}) + \Delta_{j+1}^2\tilde{\kappa}(t - t_{j+1})\right) \Delta_js(a)\Delta_j \\
&\sim_{3 L\|a\| \epsilon} \sum_{j=K}^{\infty} \tilde{\kappa}(t-t_j)\Delta_js(a)\Delta_j
\end{split}
\end{equation*}
for all large enough $t$. Similarly, we find that
$$
\left(\sum_{j=0}^{\infty}\Delta_js(a)\Delta_j\right)\left(1 -v_t\right) \sim_{4L\|a\|\epsilon} \sum_{j=K}^{\infty} \tilde{\kappa}(t-t_j)\Delta_js(a)\Delta_j
$$
for all large enough $t$. Hence (\ref{e1}) holds.

d) : We may assume that $a \in X$. Since $s$ is homogeneous, we find that
$$
\varphi_t((g \circ \kappa)\otimes a) - \left(\sum_{j=0}^{\infty} \Delta_j^2g \circ \kappa(t-t_j)\right) s(a) = \sum_{j=0}^{\infty} g \circ \kappa(t - t_j)\Delta_j \left[s(a), \Delta_j\right] .
$$
Since $\sum_{j=0}^{\infty} \left\| \left[s(a), \Delta_j\right]\right\| < \infty$, it follows easily that
$$
\lim_{t \to \infty} \left[\varphi_t((g \circ \kappa)\otimes a) - \left(\sum_{j=0}^{\infty} \Delta_j^2g \circ \kappa(t-t_j)\right) s(a)\right] = 0.
$$
By Lemma \ref{a7}, $\lambda_t(h) = \sum_{j=0}^{\infty} \Delta_j^2h \circ \kappa(t-t_j)$ gives us an asymptotic homomorphism $\lambda : C_0(0,1] \to B$. It follows that
$$
\{h \in C_0(0,1]: \ \lim_{t \to \infty} h(v_t) - \lambda_t(h) = 0\}
$$
is a $C^*$-subalgebra of $C_0(0,1]$. Since it contains the identity function it must be all of $C_0(0,1]$, so we conclude that
$$
\lim_{t \to \infty} \left[g(v_t) s(a) - \left(\sum_{j=0}^{\infty} \Delta_j^2g \circ \kappa(t-t_j)\right) s(a)\right] = 0.
$$
\end{proof}






In conclusion: Let $CH(\psi) : SA \to B$ be the asymptotic homomorphism arising from the Connes-Higson construction introduced in \cite{CH}. Then $CH(\psi)$ is homotopic to a translation invariant asymptotic homomorphism.



\section{From translation invariant asymptotic homomorphisms to extensions.}\label{Sec I}

In this section we construct an inverse of $CH_{\tau}$.





Let $\beta_0 \in C_0(\mathbb R)$ be a continuous function with support in $[-1,1]$ such that
\begin{equation}\label{e12}
\beta_0(t) > 0, \ t \in ]-1,1[,
\end{equation}\begin{equation}%\label{e10}
\tau_1(\beta_0)\tau_{-1}(\beta_0) = 0
\end{equation}
and
\begin{equation}\label{e11}
\tau_{-1}(\beta_0)\beta_0 + \beta_0^2 + \tau_{1}(\beta_0)\beta_0 = \beta_0.
\end{equation}
One choice of $\beta_0$ could be
$$
\beta_0(t) = \begin{cases} 0, & |t| \geq 1,\\ 1 + t, & t \in ]-1,0],\\ 1-t, & t \in [0,1[ . \end{cases}
$$
When $i \in \mathbb Z, \ i \neq 0$, set $\beta_i = \tau_{-i}\left(\beta_0\right)$. Then $\left\{\beta_i\right\}_{i \in \mathbb Z}$ is a partition of unity in $C_0(\mathbb R)$ such that $\beta_i$ is supported in $[-i-1,-i+1]$. Set $\alpha_i = \sqrt{\beta_i}$, and note that
\begin{equation}\label{17}
\alpha_i\alpha_j = 0, \ |i-j| \geq 2.
\end{equation}


Let $\varphi = \left(\varphi_t\right)_{t \in \mathbb R} :  SA \to B$ be an asymptotically translation invariant asymptotic homomorphism. For convenience we assume that $\varphi_t(0) = 0$ for all $t$. Then
\begin{equation}\label{ZZ2}
\varphi_i\left(\tau_{i}(\alpha_i \alpha_j)\otimes a\right) = 0 \ \text{when $|i-j| \geq 2$},
\end{equation}
for all $a \in A$. Let $\mathbb K$ be the $C^*$-algebra of compact operators on $l^2(\mathbb Z)$, and let $e_{ij}, i,j \in \mathbb Z$, be the standard matrix units in $\mathbb K$. 


Define $\Phi_0: A \to M(B \otimes \mathbb K)$ by
$$
\Phi_0(a) = \sum_{i,j \in \mathbb Z} \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_j\right)\otimes a\right) \otimes e_{ij}.
$$
The sum converges in the strict topology because of (\ref{ZZ2}) and $\sup_{i,j} \left\|\varphi_i\left( \tau_{i}\left(\alpha_i \alpha_j\right)\otimes a\right)\right\| < \infty$. Let $q_{B \otimes \mathbb K} : M(B \otimes \mathbb K) \to Q(B \otimes \mathbb K)$ be the quotient map, and set $\Phi = q_{B \otimes \mathbb K} \circ \Phi_0$.

\begin{lem}\label{l6} $\Phi$ is an extension of $A$ by $B \otimes \mathbb K$, i.e. $\Phi \in \Hom(A,Q(B \otimes \mathbb K))$.
\end{lem}
\begin{proof} Let $a,b \in A$. We check that $\Phi(a)\Phi(b) = \Phi(ab)$. Using (\ref{ZZ2}) we find that
\begin{equation}\label{e9}
\left\|\Phi(a)\Phi(b) - \Phi(ab)\right\|
\end{equation}
$$
\leq 3 \lim_{K \to \infty} \sup_{|i| \geq K,|k -i| \leq 3} \left\|\varphi_i\left(\tau_{i}(\alpha_i \alpha_k)\otimes ab\right) - \sum_{j \in \mathbb Z} \varphi_i\left( \tau_{i}(\alpha_i\alpha_j)\otimes a \right)  \varphi_j\left( \tau_{j}(\alpha_j\alpha_k)\otimes b \right) \right\| .
$$
Using (\ref{ZZ2}) again we find that
\begin{equation*}
\begin{split}
&\sum_{j \in \mathbb Z} \varphi_i\left(\tau_{i}(\alpha_i \alpha_j)\otimes a\right)\varphi_j\left(\tau_{j}(\alpha_j \alpha_k)\otimes b\right) \\
&=  \varphi_i\left(\tau_{i}(\alpha_i^2)\otimes a\right) \varphi_i\left(\tau_{i}(\alpha_i \alpha_k)\otimes b\right) + \varphi_i\left(\tau_{i}(\alpha_i\alpha_{i-1})\otimes a\right) \varphi_{i-1}\left(\tau_{i-1}(\alpha_{i-1} \alpha_k)\otimes b\right) \\
&+ \varphi_i\left(\tau_{i}(\alpha_i\alpha_{i+1})\otimes a\right) \varphi_{i+1}\left(\tau_{i+1}(\alpha_{i+1} \alpha_k)\otimes b\right)\\
& = \varphi_i\left(\alpha_0^2\otimes a\right) \varphi_i\left(\alpha_0 \alpha_{k-i}\otimes b\right) + \varphi_i\left(\alpha_0\alpha_{-1}\otimes a\right) \varphi_{i-1}\left(\alpha_{0} \alpha_{k-i+1}\otimes b\right) \\
&+ \varphi_i\left(\alpha_0\alpha_{1}\otimes a\right) \varphi_{i+1}\left(\alpha_{0} \alpha_{k-i-1}\otimes b\right).
\end{split}
\end{equation*}
Since $\varphi$ is translation invariant, we have that
$$
\lim_{i \to \infty} \sup_{|k-i| \leq 3} \left\|\varphi_i\left(\alpha_0\alpha_{-1}\otimes a\right) \varphi_{i-1}\left(\alpha_{0} \alpha_{k-i+1}\otimes b\right) - \varphi_i\left(\alpha_0\alpha_{-1}\otimes a\right) \varphi_{i}\left(\alpha_{-1} \alpha_{k-i}\otimes b\right)\right\|  = 0
$$
and
$$
\lim_{i \to \infty} \sup_{|k-i| \leq 3}\left\| \varphi_i\left(\alpha_0\alpha_{1}\otimes a\right) \varphi_{i+1}\left(\alpha_{0} \alpha_{k-i-1}\otimes b\right) -  \varphi_i\left(\alpha_0\alpha_{1}\otimes a\right) \varphi_{i}\left(\alpha_{1} \alpha_{k-i}\otimes b\right)\right\| = 0 .
$$

It follows that
\begin{eqnarray*}
&&\lim_{i \to \infty} \sup_{|k-i|\leq 3} \Bigl\|\sum_{j \in \mathbb Z} \varphi_i\left(\tau_{i}(\alpha_i \alpha_j)\otimes a\right)\varphi_j\left(\tau_{j}(\alpha_j \alpha_k)\otimes b\right)\Bigr. \\
&&\ \ \ \ \ \ \ \ \ \ \ \ \ \ \Bigl.- \varphi_i\left(\left(\alpha_0\alpha_0^2\alpha_{k-i} + \alpha_0 \alpha_{-1}^2\alpha_{k-i}+ \alpha_0 \alpha_{1}^2\alpha_{k-i}\right)\otimes ab \right)\Bigr\| = 0.
\end{eqnarray*}
Since $\alpha_0\alpha_0^2\alpha_{k-i} + \alpha_0 \alpha_{-1}^2\alpha_{k-i}+ \alpha_0 \alpha_{1}^2\alpha_{k-i} = \tau_{i}\left(\alpha_i\alpha_k\right)$ by (\ref{e11}), we see that
\begin{equation}\label{e8}
\lim_{i \to \infty} \sup_{|k-i|\leq 3} \left\|\sum_{j \in \mathbb Z} \varphi_i\left(\tau_{i}(\alpha_i \alpha_j)\otimes a\right)\varphi_j\left(\tau_{j}(\alpha_j \alpha_k)\otimes b\right) - \varphi_i\left(\tau_{i}(\alpha_i \alpha_k)\otimes ab\right)\right\| = 0.
\end{equation}
By using that $\lim_{t \to -\infty} \varphi_t(f) = 0$ for all $f \in SA$, we find easily that
\begin{equation}\label{e10}
\begin{split}
&
\lim_{i \to -\infty} \sup_{|k-i| \leq 3} \left\|\sum_{j \in \mathbb Z} \varphi_i\left(\tau_{i}(\alpha_i \alpha_j)\otimes a\right)\varphi_j\left(\tau_{j}(\alpha_j \alpha_k)\otimes b\right)\right\| \\
& = \lim_{i \to -\infty} \sup_{|k-i| \leq 3} \left\|\varphi_i\left(\tau_{i}(\alpha_i \alpha_k)\otimes ab\right)\right\| = 0.
\end{split}
\end{equation}
By combining (\ref{e9}), (\ref{e8}) and (\ref{e10}), we conclude that $\Phi(a)\Phi(b) = \Phi(ab)$.  It can be shown in a similar way that $\Phi$ is linear and self-adjoint. Thus $\Phi$ is an extension of $A$ by $B \otimes \mathbb K$.
\end{proof}



The extension $\Phi$ of Lemma \ref{l6} will be denoted by $I(\varphi)$. It is clear that the construction gives rise to a well-defined map
$$
I : [[SA,B]]_{a, \tau} \to \Ext_h(A,B \otimes \mathbb K)
$$
such that $I[\varphi] = [I(\varphi)]$.



\begin{rmk}\label{remork} \emph{At first sight it may seem odd that $I(\varphi)$ only depends of the 'discrete part' of the asymptotic homomorphism $\varphi$; i.e. of $\varphi_z, z \in \mathbb Z$. This is not so strange because a discrete translation invariant asymptotic homomorphism $\psi$, as defined in the obvious way, determines a unique element in $[[SA,B]]_{a,\tau}$ via the formula
$$
\psi_t = \psi_{[t]} \circ \tau_{[t] -t} .
$$}    
\end{rmk}













\begin{rmk}\label{remark} \emph{We want to point out that the $I$ construction, considered as a map $[[SA,B]]_{a, \tau} \to \Ext_h(A,B \otimes \mathbb K)$, can be wrapped in other, maybe more familiar guises: As pointed out in Remark \ref{remork} there is nothing lost in thinking of $\varphi$ as a translation invariant \emph{discrete} asymptotic homomorphism, and then the discrete version of the $M$-construction from \cite{H-LT} gives us a $\mathbb Z$-extension $E$ of $SA$ by $C_0(\mathbb Z,B)$. The crossed product $E \times \mathbb Z$ is then an extension of  $C(\mathbb T)\otimes \mathbb K  \otimes A$ by $B \otimes \mathbb K$. The pull back of this extension along the homomorphism $a \mapsto e \otimes a$, where $e \in C(\mathbb T)\otimes \mathbb K$ is a projection which generates $K_0( C(\mathbb T)\otimes \mathbb K) \simeq \mathbb Z$, gives us an  extension of $A$ by $B \otimes \mathbb K$ which is homotopic to $I(\varphi)$. Alternatively, the $I$ map can be realized via the genuine M-construction, using crossed products by $\mathbb R$.} 
\end{rmk} 



\section{The main result}

In this section we prove our main result by showing that $I$ and $CH_{\tau}$ are the inverses of each other when $B$ is stable.

\begin{lem}\label{l10} Let $s : B \to B \otimes \mathbb K$ be the stabilizing $*$-homomorphism, $s(b) = b \otimes e_{00}$, and let $\hat{s} : Q(B) \to Q(B \otimes \mathbb K)$ be the embedding induced by $s$. Define $s_* : \Ext_h(A,B) \to  \Ext_h(A, B \otimes \mathbb K)$ such that $s_*[\varphi] = [\hat{s} \circ \varphi]$. Then the diagram
\begin{equation*}
\begin{xymatrix}{
\Ext_h(A,B) \ar[r]^-{s_*} \ar[d]_-{CH_{\tau}} & \Ext_h(A, B \otimes \mathbb K) \\
[[SA,B]]_{a, \tau} \ar[ur]_-{I} & }
\end{xymatrix}
\end{equation*}
commutes.
\end{lem}
\begin{proof} We adopt the notation from above and from Section \ref{vch}. Let $\varphi : A \to Q(B)$ be an extension. Then $I \circ CH_{\tau}[\varphi]$ is represented by the map
$$
\Phi(a) = q_{B \otimes \mathbb K}\left( \sum_{i,j \in \mathbb Z} \sum_{k =0}^{\infty} \Delta_k^2 \alpha_i\alpha_j(-t_k) s(a) \otimes e_{ij} \right) .
$$
We may assume that there is a compact subset $X \subseteq A$ with dense span in $A$ such that
\begin{equation}\label{e15}
\sum_{k=0}^{\infty} \left\|\left[ \Delta_j, s(a)\right]\right\| < \infty
\end{equation}
for all $a \in X$, cf. Lemma \ref{a5}. Let $\lambda : I \to B$ be the completely positive asymptotic homomorphism $\lambda_t(h) = \sum_{j=0}^{\infty} \Delta_j^2 h(t-t_j)$, cf. Lemma \ref{a7}. Then
$$
 q_{B \otimes \mathbb K}\left( \sum_{i,j \in \mathbb Z} \sum_{k =0}^{\infty} \Delta_k^2 \alpha_i\alpha_j(-t_k) s(a) \otimes e_{ij} \right) =  q_{B \otimes \mathbb K} \left( \sum_{i,j \in \mathbb Z} \lambda_0(\alpha_i\alpha_j) s(a) \otimes e_{ij} \right) .
$$
Since $\lim_{k \to \infty} t_k - t_{k+1} = 0$, there is an $N \in \mathbb N$ so large that $\left|t_k - t_{k+1}\right| \leq N$ for all $k$. Since $\alpha_0$ is supported in $[-1,1]$, it follows from (\ref{e13}) that
$$
\lambda_i(\alpha_0)\lambda_j(\alpha_0) = \sum_{k,l =0}^{\infty} \Delta_k^2\Delta_l^2 \alpha_0(i-t_k)\alpha_0(j-t_l) = 0
$$
when $|i -j| \geq N+3$. Hence
$$
\sup_j \left\|\lambda_0(\alpha_i\alpha_j) - \lambda_0(\alpha_i)\lambda_0(\alpha_j)\right\| = \sup_j \left\|\lambda_i(\alpha_0\alpha_{j-i}) - \lambda_i(\alpha_0)\lambda_i(\alpha_{j-i})\right\|
$$
tends to $0$ as $i$ tends to $\pm \infty$. It follows that
$$
 q_{B \otimes \mathbb K}\left( \sum_{i,j \in \mathbb Z} \lambda_0(\alpha_i\alpha_j) s(a) \otimes e_{ij} \right)
=  q_{B \otimes \mathbb K}\left( \sum_{i,j \in \mathbb Z} \lambda_i(\alpha_0)\lambda_j(\alpha_0) s(a) \otimes e_{ij} \right).
$$
Set
$$
V = \sum_{i \in \mathbb Z} \lambda_0(\alpha_i) \otimes e_{0i} \in M(B \otimes \mathbb K),
$$
where the sum converges in the strict topology. By using that $\sum_{i \in \mathbb Z}\lambda_i(\alpha_0)^2 = 1$, with convergence in the strict topology, we see that $VV^* = 1 \otimes e_{00} \in M(B \otimes \mathbb K)$. Note that
$$
V^*\left(s(a) \otimes e_{00} \right) V =  \sum_{i,j \in \mathbb Z} \lambda_i(\alpha_0)s(a)\lambda_j(\alpha_0)  \otimes e_{ij} .
$$
It follows from (\ref{e15}) that $\lim_{i \to \infty} \lambda_i(\alpha_0)s(a) - s(a)\lambda_i(\alpha_0) = 0$ for all $a \in X$. Since $\lim_{i \to -\infty} \lambda_i(\alpha_0) = 0$, we conclude that
$$
V^* \left(s(a) \otimes e_{00} \right)V  =   \sum_{i,j \in \mathbb Z} \lambda_i(\alpha_0)\lambda_j(\alpha_0) s(a) \otimes e_{ij} 
$$
modulo $B \otimes \mathbb K$ for all $a \in X$. Then $U = \left( \begin{smallmatrix} V & 1-VV^*\\ 1-V^*V & V^* \end{smallmatrix} \right) \in M_2(M(B \otimes \mathbb K))$ is a unitary such that
\begin{equation}\label{e16}
q_{M_2(B \otimes \mathbb K)} \left( \Ad U \circ  \left( \begin{smallmatrix} \Phi(a) & \\  & 0 \end{smallmatrix} \right)\right)  =q_{M_2(B \otimes \mathbb K)}  \left( \begin{smallmatrix} s(a) \otimes e_{00} &  \\  & 0 \end{smallmatrix} \right)
\end{equation}
for all $a \in X$. Since $X$ spans a dense set in $A$, (\ref{e16}) holds for all $a \in A$. Now the lemma follows because $U$ can be connected to $1$ in the strict topology, by a path of unitaries in $M(M_2(B \otimes \mathbb K))$.
\end{proof}


\begin{lem}\label{Lem2} Let $\varphi = \left( \varphi_t\right)_{t \in \mathbb R} : SA \to B$ be an asymptotically translation invariant asymptotic homomorphism. It follows that there is a homogeneous equi-continuous and uniformly continuous asymptotic homomorphism $\psi =  (\psi_t)_{t \in \mathbb R} : SA \to B$ such that
\begin{enumerate}
\item[1)] $\lim_{t \to \infty} \psi_t(f) - \varphi_t(f) = 0$ for all $f \in SA$,
\item[2)] $\lim_{t \to \infty} \sup_{s \in K} \left\|\psi_{t-s}(f) - \psi_t\left(\tau_s(f)\right)\right\| = 0$ for all $f \in SA$ and all compact subsets $K \subseteq \mathbb R$,
\item[3)] $\lim_{t \to -\infty} \psi_t(f) = 0$ for all $f \in SA$.
\end{enumerate}

\end{lem}
\begin{proof} Let $X = \left\{ f \in C_b(\mathbb R,B) : \ \lim_{t \to - \infty} f(t) = 0\right\}$ and $X_0 = \left\{f \in X: \ \lim_{t \to \infty} f(t) = 0\right\}$. Define an automorphism $\theta_x$ of $X$ such that $\theta_x(f)(t) = f(t-x)$, and note that $\theta_x$ leaves $X_0$ globally invariant, so that $\theta_x$ induces an automorphism of $X/X_0$ which we denote by $\theta_x$ again. Let $\Phi_0 : SA \to X$ be the map $\Phi_0(f)(t) = \varphi_t(f)$. Let $q : X \to X/X_0$ 
be the quotient map, and note that $\Phi = q \circ \Phi_0$ is then an equivariant $*$-homomorphism. Let $X_{\mathbb R}$ denote the $C^*$-subalgebra of $X$ consisting of the elements $f \in X$ for which $x \mapsto \theta_x(f)$ is continuous. It follows from Theorem 2.1 of \cite{Th?} that $ \Phi(SA) \subseteq q\left(X_{\mathbb R}\right)$. Let $L : q\left( X_{\mathbb R}\right)  \to X_{\mathbb R}$ be a continuous and 
homogeneous right-inverse for $q : X_{\mathbb R} \to q\left( X_{\mathbb R}\right)$, and set $\psi_t(f) = L \circ \Phi(f)(t)$. Then $\psi = \left( \psi_t\right)_{t \in \mathbb R}$ is an equicontinuous and homogeneous asymptotic homomorphism such that $\lim_{t \to \infty} \psi_t(f) - \varphi_t(f) = 0$, $\lim_{t \to - \infty} \psi_t(f) = 0$ and
\begin{equation}\label{16}
\lim_{t \to \infty} \psi_t\left(\tau_s(f)\right) - \psi_{t-s}(f) = 0
\end{equation}
for all $f \in SA$ and all $s \in \mathbb R$. The uniform continuity of $\psi$ follows because $L \circ \Phi(f) \in X_{\mathbb R}$. And then 2) follows from (\ref{16}), the uniform continuity and the equicontinuity of $\psi$, as follows: By uniform continuity there is $\delta > 0$ so small that $\sup_{t \in \mathbb R}\|\psi_{t-x}(f) - \psi_{t-y}(f)\| \leq \epsilon$ when $|x-y| < \delta$. By equicontinuity of $\psi$ there is for each $s \in K$, a $\delta_s \in ] 0, \delta[$ such 
that $\sup_{t \in \mathbb R} \left\|\psi_{t}(\tau_s(f)) - \psi_t(\tau_{s'}(f))\right\| < \epsilon$ when $|s'-s| < \delta_s$. Let $\left]s_i - \delta_{s_i}, s_i + \delta_{s_i}\right[, i = 1,2, \dots, L$, be a finite subcover of the cover $]s - \delta_s,s + \delta_s[, s \in K$, of $K$. Choose $T > 0$ so large that $\left\|\psi_{t-s_i}(f) - \psi_t \circ \tau_{s_i}(f)\right\| < \epsilon$ for all $i = 1,2, \dots, L$, when $t > T$. It follows then that
$$
 \sup_{s \in K} \left\|\psi_{t-s}(f) - \psi_t\left(\tau_s(f)\right)\right\| < 3 \epsilon
$$
when $t > T$.
\end{proof}



It follows from Lemma \ref{Lem2} that we can always replace an asymptotically translation invariant asymptotic homomorphism with one which has the properties spelled out in Lemma \ref{Lem2}; namely, uniform continuity and equicontinuity. Property 2) of Lemma \ref{Lem2} is then automatic.





\begin{lem}\label{pre} Let $\varphi = \left( \varphi_t\right)_{t \in \mathbb R} : SA \to B$ be an equi-continuous asymptotically translation invariant asymptotic homomorphism, and let $M \subseteq SA$ be a pre-compact subset. It follows that
\begin{enumerate}
\item[i)] $\lim_{t \to - \infty} \sup_{x \in M} \left\|\varphi_t(x)\right\| = 0$.
\item[ii)] For all $\epsilon > 0$, there is a $T > 0$ such that
$$
\left\|\varphi_t(x) -\varphi_t(y)\right\| \leq \| x-y\| + \epsilon
$$
for all $x,y \in M$ and all $t \geq T$.
\item[iii)]  For all $\epsilon > 0$, there is a $T > 0$ such that
$$
\left\|\varphi_t(x)\varphi_t(y) - \varphi_t(xy)\right\| \leq \epsilon
$$
for all $x,y \in M$ and all $t \geq T$.
\item[iv)]  For all $\epsilon > 0$, there is a $T > 0$ such that
$$
\left\|\varphi_t(x) + \varphi_t(y) - \varphi_t(x+y)\right\| \leq \epsilon
$$
for all $x,y \in M$ and all $t \geq T$.
\item[v)]  For any $\epsilon > 0$, and any compact subset $K \subseteq \mathbb R$, there is a $T > 0$ such that
$$
\sup_{s \in K} \left\|\varphi_{t-s}(x) - \varphi_t(\tau_s(x)))\right\| \leq \epsilon
$$
for all $x\in M$ and all $t \geq T$.
\end{enumerate}
\end{lem}
\begin{proof} 
Left to the reader.
\end{proof}


\begin{lem}\label{l12}  The diagrams
\begin{equation*}
\begin{xymatrix}{
& \Ext_h(A, B \otimes \mathbb K) \ar[d]^-{CH_{\tau}} \\
[[SA,B]]_{a, \tau} \ar[ur]^-I  \ar[r]_-{s_*} & [[SA,B \otimes \mathbb K]]_{a, \tau} }
\end{xymatrix}
\end{equation*}
and
\begin{equation*}
\begin{xymatrix}{
& \Ext_h(A, B \otimes \mathbb K) \ar[d]^-{CH_{\tau}} \\
[[SA,B]]_{ \tau} \ar[ur]^-I  \ar[r]_-{s_*} & [[SA,B \otimes \mathbb K]]_{ \tau} }
\end{xymatrix}
\end{equation*}
both commute.
\end{lem}
\begin{proof}  As in the previous proof we will adopt the notation from above and from Section \ref{vch}. Let $\varphi = \left(\varphi_t\right)_{t \in \mathbb R} : SA \to B$ be an asymptotically translation invariant asymptotic homomorphism. By Lemma \ref{Lem2} we may assume that $\varphi$ has the properties spelled out there (for $\psi$). Set $\Lambda(a) = \sum_{i,j \in \mathbb Z} \varphi_i\left(\tau_i\left(\alpha_i\alpha_j\right)\otimes a\right) \otimes e_{ij}$. Let $c$ be a strictly positive element in $B$. Then
$$
b = \sum_{i \in \mathbb Z} 2^{-|i|}c \otimes e_{ii}
$$
is a strictly positive element of $B \otimes \mathbb K$ which we may use to construct $CH_{\tau}\left(I(\varphi)\right)$. However, we consider instead
$$
b_0 = \sum_{i \in \mathbb Z} 2^{-|i|} \otimes e_{ii},
$$
which in general is not an element of $B \otimes \mathbb K$, only of $M(B \otimes \mathbb K)$. But $b_0$ has the following two properties: $b_0\Phi_0(A) \subseteq B \otimes \mathbb K$ and $\lim_{k \to \infty} b_0^{\frac{1}{k}}x = x$ for all $x \in B \otimes \mathbb K$. We can therefore find a sequence $h_0 \leq h_1 \leq h_2 \leq \dots $ of functions in $C_0]0,1]$ such that 
\begin{equation}\label{u1}
h_{n+1}h_n = h_n
\end{equation} 
for all $n$, and the sequence $u_n = h_n(b_0), n = 0,1,2, \dots$ has the properties 
\begin{enumerate}
 \item[a)] $\lim_{n \to \infty} u_n\Phi_0(a) - \Phi_0(a)u_n =0$ for all $a \in A$, and \item[b)] $\lim_{n \to \infty} u_nx = x$ for all $x \in B \otimes \mathbb K$. 
\end{enumerate}
Set $\Delta_0 = \sqrt{u_0}, \Delta_n = \sqrt{u_n - u_{n-1}}, n \geq 1$. The arguments from Section \ref{subsec} show that $CH_{\tau} \circ I[\varphi]$ is represented by a translation invariant asymptotic homomorphism $\psi : SA \to B \otimes \mathbb K$ such that
\begin{equation}\label{u3}
\lim_{t \to \infty} \left[\psi_t(f \otimes a) - \Phi_0(a)\sum_{j=0}^{\infty} \Delta_j^2 f(t-t_j)\right] = 0 .
\end{equation}
Note that we can choose the sequence $\{t_j\}_{j=0}^{\infty}$ in $[0,\infty)$ at will, as long as $\lim_{j \to \infty} t_j = \infty$ and $\lim_{j \to \infty} t_{j+1} - t_j = 0$. This freedom is used as follows: For each $i \in 0,1,2,3, \dots $, set
$$
j_i = \min \left\{ k : h_k\left(2^{-i}\right) = 1 \right\} .
$$
It follows from (\ref{u1}) that $j_i \leq j_{i+1}$ and from b) that $\lim_{i \to \infty} j_i = \infty$. We can therefore choose $\{t_j\}_{j=0}^{\infty}$ such that $\lim_{i \to \infty} t_{j_{i}} - t_{j_{i-1}} = 0$. With this choice, it follows from (\ref{u1}) that
\begin{equation}\label{u2}
\begin{split}
&\sum_{j=0}^{\infty} \Delta_j^2 f(t-t_j) = \\
& \sum_{i \in \mathbb Z} \left(\left(1-h_{j_{|i|} -1}\left(2^{-|i|}\right)\right) f\left(t-t_{j_{|i|}}\right) +  h_{j_{|i|} -1}\left(2^{-|i|}\right)f\left(t - t_{j_{|i|} -1}    \right) \right) e_{ii} ,
\end{split}
\end{equation}
if we set $h_{-1} = 0$. Set $s_i = t_{j_{|i|}}, i \in \mathbb Z$, and note that $\lim_{i \to \pm \infty} \left|s_i - s_{i-1}\right| = 0$ and $\lim_{i \to \pm \infty} s_i = \infty$. It follows from (\ref{u2}) that
$$
\lim_{t \to  \infty} \left[\sum_{j=0}^{\infty} \Delta_j^2 f(t-t_j) - \sum_{i \in \mathbb Z} f(t-s_i)e_{ii}\right] = 0
$$
for all $f \in C_0(\mathbb R)$, and then from (\ref{u3}) that
$$
\lim_{t \to \infty} \left[\psi_t(f \otimes a) -  \sum_{i,j \in \mathbb Z} \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_jf(t-s_i )\right)\otimes a\right) \otimes e_{ij}\right] = 0,
$$
for all $f \in C_0(\mathbb R), a \in A$. Let $\beta : \mathbb R \to \mathbb R$ be a continuously differentiable decreasing function such that $\beta(i) = s_i =t_{j_{|i|}}$ for all $i\in \mathbb Z$, $i < 0$, while $\beta(x) = -x, x \geq 0$, and such that $\beta^{\prime}$ is bounded. The last condition can be met because  $\lim_{i \to \pm \infty} \left|s_i - s_{i-1}\right| = 0$. Then
\begin{equation}\label{u5}
\lim_{t \to \infty} \left[\psi_t(f \otimes a) -  \sum_{i,j \in \mathbb Z} \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_jf( t -\beta(\cdot))\right)\otimes a\right) \otimes e_{ij}\right] = 0.
\end{equation}
Indeed, we claim that
\begin{equation}\label{u4}
\lim_{t \to \infty} \sup_{i,j \in \mathbb Z} \left\| \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_jf(t-s_i )\right)\otimes a\right) - \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_jf( t -\beta(\cdot))\right)\otimes a\right)\right\| = 0.
\end{equation}
To see this, let $\epsilon > 0$ and observe that $\tau_{i}\left(\alpha_i \alpha_jf(t-s_i )\right) = \alpha_0\alpha_{j-i} f(t-s_i)$ is a pre-compact set in $C_0(\mathbb R)$. By i) of Lemma \ref{pre} there is a $K < 0$ such that 
\begin{equation}\label{u8}
 \left\| \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_jf(t-s_i )\right)\otimes a\right)\right\| \leq \epsilon
\end{equation}
for all $i \leq K$. Similarly, it follows from Ascoli's theorem, cf. e.g. Theorem A 5 on page 369 of \cite{R}, that  
$$
\left\{ \tau_{i}\left(\alpha_i \alpha_jf( t -\beta(\cdot))\right) : i \in \mathbb Z, t \in \mathbb R \right\} = \left\{ \alpha_0 \alpha_{j-i}f( t -\beta(\cdot - i)) : i \in \mathbb Z, t \in \mathbb R \right\}
$$
is pre-compact in $C_0(\mathbb R)$. By using i) of Lemma \ref{pre} again, we may assume, by decreasing $K$ if necessary, that also  
\begin{equation}\label{u9}
\left\| \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_jf(t- \beta(\cdot ) )\right)\otimes a\right)\right\| \leq \epsilon 
\end{equation}
 for all $i \leq K$. It follows from the uniform continuity of $f$, and the fact that $\lim_{i \to \infty} \sup_{x \in [-1,1]} |s_i -\beta(x - i)| = 0$, that
$$
\lim_{i \to \infty} \sup_{t,j} \left\| \alpha_0\alpha_{j-i}f(t - s_i) -  \alpha_0\alpha_{j-i}f(t - \beta(\cdot - i))\right\| = 0.
$$
It follows therefore from ii) of Lemma \ref{pre} that there is an $L > 0$ so large that
\begin{equation}\label{u6}
\left\| \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_jf(t-s_i )\right)\otimes a\right) - \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_jf( t -\beta(\cdot))\right)\otimes a\right)\right\| \leq \epsilon
\end{equation}
when $i \geq L$. Since 
$$
\lim_{t \to \infty} \sup_{K \leq i\leq L} \left\| \tau_{i}\left(\alpha_i \alpha_jf(t-s_i )\right)\right\| = \lim_{t \to \infty}  \sup_{K \leq i\leq L} \left\|\tau_{i}\left(\alpha_i \alpha_jf( t -\beta(\cdot))\right)\right\| = 0,
$$ 
we conclude from (\ref{u8}),(\ref{u9}) and (\ref{u6}) that (\ref{u4}) holds. Hence (\ref{u5}) holds. 


To proceed it is useful to apply the following statement, which is an abstract version of Ascoli's theorem. In fact, the proof is identical to the standard proof of Ascoli's theorem, as presented for example in \cite{R}. Alternatively, it can be deduced from the version of Ascoli's theorem on page 81 of \cite{Ke}. 



\begin{thm}\label{aA} Let $K$ be a locally compact Hausdorff space, $X$ a Banach space, and $C_0(K,X)$ the Banach space of continuous $X$-valued functions on $K$ that vanish at infinity, with the norm $\|f\| = \sup \left\{\|f(k)\|: \ k \in K\right\}$. A subset $M \subseteq C_0(K,X)$ is pre-compact (i.e. $\overline{M}$ is compact) in $C_0(K,X)$ if and only if
\begin{enumerate}
\item[a)] $\overline{ \{f(k): f \in M\}}$ is compact in $X$ for all $k \in X$,
\item[b)] $M$ is equi-continuous, i.e. for all $\epsilon > 0$, every $k \in K$ has a neighbourhood $V$ such that $\|f(k) - f(k')\| < \epsilon$ for all $k' \in V$ and all $f \in M$,
\item[c)] the elements of $M$ vanish equicontinuously at infinity, i.e. for all $\epsilon > 0$ there is compact subset $C \subseteq K$ such that $\|f(k)\| \leq \epsilon$ for all $f \in M$ and all $k \in K \backslash C$.
\end{enumerate}
\end{thm}






























Set now $\beta_{\lambda}(x) = \lambda \beta(x) - (1-\lambda)x$, and note that $\sup_{\lambda \in [0,1]} \left\|\beta_{\lambda}^{\prime}\right\| < \infty$. It follows then from Theorem \ref{aA} that for any $f \in SA = C_0(\mathbb R, A)$, the set 
$$
\left\{ \alpha_0\alpha_{j-i}f\left(t - \beta_{\lambda}( \cdot - i)\right): \ j,i \in \mathbb Z, t \in \mathbb R, \lambda \in [0,1] \right\}
$$
is a pre-compact set in $SA$. We claim that as a consequence of this we have that
\begin{align}\label{uu1}
\lim_{t \to \infty} \sup_{i,k \in \mathbb Z} \sup_{\lambda \in [0,1]} \bigg\|\sum_{j \in
    \mathbb Z}  \varphi_i\left( \tau_{i}\left(\alpha_i
      \alpha_jf(t - \beta_{\lambda}(\cdot))\right)\right)  \varphi_j\left(
    \tau_{j}\left(\alpha_j \alpha_kg(t -\beta_{\lambda}(\cdot))\right)\right)
  \\-  \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_kfg(t -\beta_{\lambda}(\cdot)
   )\right)\right)\bigg\| = 0
\end{align}
for all $f,g \in SA$. 


To see this, note that by iii) and iv) of Lemma \ref{pre}, there is a $K > 0$ such that
\begin{align}\label{uu2}
  \bigg\|\sum_{j \in
    \mathbb Z}  \varphi_i\left( \tau_{i}\left(\alpha_i
      \alpha_jf(t - \beta_{\lambda}(\cdot))\right)\right)  \varphi_j\left(
    \tau_{j}\left(\alpha_j \alpha_kg(t -\beta_{\lambda}(\cdot))\right)\right)
  \\-  \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_kfg(t -\beta_{\lambda}(\cdot)
   )\right)\right)\bigg\| \leq \epsilon
\end{align}
for all $\lambda \in [0,1],k \in \mathbb Z$ when $i \geq K$. Since $\epsilon > 0$ is arbitrary, (\ref{uu1}) will follow if we can show that there is a $T > 0$ so large that 
\begin{equation}\label{uu3}
 \left\|\varphi_i\left( \tau_{i}\left(\alpha_i
      \alpha_jf(t - \beta_{\lambda}(\cdot))\right)\right)\right\| \leq \epsilon
\end{equation}    
for all $i \leq K+1$ and all $\lambda \in [0,1]$ when $t \geq T$. Since 
\begin{equation}\label{uu14}
\left\{ \alpha_0 \alpha_{j-i} f(t - \beta_{\lambda} ( \cdot - i)) : \ i,j \in \mathbb Z, t \in \mathbb R, \lambda \in [0,1]\right\}
\end{equation}
is a pre-compact set in $SA$ by Theorem \ref{aA}, it follows from i) of Lemma \ref{pre} that there is a $L < 0$ such that (\ref{uu3}) holds for all $t \in \mathbb R$ and all $\lambda \in [0,1]$ when $i \leq L$. Since
$$
\lim_{t \to \infty} \sup \left\{ \left\| \alpha_0(s) \alpha_{j-i}(s) f(t - \beta_{\lambda} ( s  - i)) \right\|  : \ s \in [-1,1], L \leq i \leq K \right\} = 0,
$$
the equi-continuity of $\varphi$ at $0$ ensures that 
$$
\lim_{ t \to \infty} \sup \left\{\left\|\varphi_i\left( \tau_{i}\left(\alpha_i
      \alpha_jf(t - \beta_{\lambda}(\cdot))\right)\right)\right\|:  \ \lambda \in [0,1], \ L  \leq i \leq K \right\} = 0.
$$
We can therefore find $T > 0$ such that (\ref{uu3}) holds, proving (\ref{uu1}). Similar arguments show that 
\begin{align}\label{uu11}
\lim_{t \to \infty} \sup_{i,j \in \mathbb Z} \sup_{\lambda \in [0,1]} \bigg\|  \varphi_i\left( \tau_{i}\left(\alpha_i
      \alpha_jf(t - \beta_{\lambda}(\cdot))\right)\right)  + \mu \varphi_i\left(
    \tau_{i}\left(\alpha_i \alpha_jg(t -\beta_{\lambda}(\cdot))\right)\right)
  \\-  \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_j\left(f + \mu g\right)(t -\beta_{\lambda}(\cdot)
   )\right)\right)\bigg\| = 0,
\end{align}
and that
\begin{equation}\label{uu12}
\lim_{t \to \infty} \sup_{i,j \in \mathbb Z} \sup_{\lambda \in [0,1]} \bigg\|  \varphi_i\left( \tau_{i}\left(\alpha_i
      \alpha_jf(t - \beta_{\lambda}(\cdot))^*\right)\right) -  \varphi_j\left( \tau_{j}\left(\alpha_i \alpha_jf(t -\beta_{\lambda}(\cdot)
   )\right)\right)^*\bigg\| = 0,
\end{equation}
for all $f,g \in SA, \mu \in \mathbb C$. For $t \in \mathbb R$, define $\Phi_t : SA \to IB$ by
$$
\Phi_t(f)(\lambda) =  \sum_{i,j \in \mathbb Z} \varphi_i\left( \tau_{i}\left(\alpha_i \alpha_jf( t -\beta_{\lambda}(\cdot))\right)\right) \otimes e_{ij} .
$$
The continuity in $\lambda$ follows from another application of Theorem \ref{aA} and ii) of Lemma \ref{pre}. To see that that $\lim_{t \to - \infty} \Phi_t(f) = 0$ use first the pre-compactness of (\ref{uu14}) to find a $K < 0$ such that 
\begin{equation}\label{ZZ3}
\sup_{j,t,\lambda} \left\|\varphi_i\left( \tau_i (\alpha_i \alpha_j f(t - \beta_{\lambda}( \cdot)) )\right)\right\| \leq \epsilon
\end{equation} 
when $i \leq K$. Then observe that there is a $c \in \mathbb R$ such that $\beta_{\lambda}(x - i) \geq c$ for all $\lambda \in [0,1]$, all $x \in [-1,1]$ and all $i \geq K$. It follows that 
$$
\lim_{t \to -\infty} \left[\sup_{\lambda, i \geq K} \left\| \alpha_0\alpha_{j-i}f\left(t - \beta_{\lambda}( \cdot - i)\right)\right\| \right] = 0.
$$ 
The equi-continuity of $\varphi$ at $0$ implies then that
$$
\lim_{t \to -\infty} \left[\sup_{\lambda, i \geq K} \left\| \varphi_i \left( \tau_i \left(\alpha_i\alpha_{j}f\left(t - \beta_{\lambda}( \cdot)\right)\right)\right)\right\| \right] = 0.
$$     
Combined with (\ref{ZZ3}) this shows that $\lim_{t \to - \infty} \Phi_t(f) = 0$, as desired. Together with (\ref{uu1}), (\ref{uu11}) and (\ref{uu12}) we see that $\left(\Phi_t\right)_{t \in \mathbb R} : SA \to IB$ is an asymptotic homomorphism. We claim that $\Phi$ is translation invariant. The algebraic condition, that $\Phi_{t-s} = \Phi_t \circ \tau_s$, is trivially satisfied, so it remains to show that $\Phi$ is equi-continuous. To prove this, let $C \subseteq SA$ be a compact subset containing the set (\ref{uu14}). Let $\epsilon > 0$. By equi-continuity of $\varphi$ and compactness of $C$ there is a finite set $g_1,g_2, \dots, g_N \in C$ and a $\delta > 0$ such that $\bigcup_i \{h \in SA: \ \|h - g_i\| < \frac{\delta}{2}\} \supseteq C$, and $\sup_t \|\varphi_t(h) - \varphi_t(g_i)\| \leq \epsilon$ when $\|h - g_i\| < \delta$. It follows $\sup_t \|\Phi_t(h) - \Phi_t(f)\| \leq  3\epsilon$ when $\|h - f\| \leq \frac{\delta}{2}$, proving that $\Phi$ is indeed equi-continuous. Thus $\Phi$ gives us a homotopy of translation invariant asymptotic homomorphism which thanks to (\ref{u5}) connects $\psi$ with the translation invariant asymptotic homomorphism $\psi' : SA \to B$ such that
$$
\psi'_t(f) = \sum_{i,j \in \mathbb Z} \varphi_i \left( \tau_i \left( \alpha_i \alpha_j f(t + \cdot)\right)\right) \otimes e_{ij},
$$
$f \in SA$. Note that 
$$
\sum_{i,j \in \mathbb Z} \varphi_i \left( \tau_i \left( \alpha_i \alpha_j f(t + \cdot)\right)\right) \otimes e_{ij} = \sum_{i,j \in \mathbb Z} \varphi_{t + (i-t)} \left(  \alpha_0 \alpha_{j-i} f(t + \cdot - i)\right) \otimes e_{ij}.
$$
Let $\epsilon > 0$. Since $\varphi$ is equi-continuous at $0$ there is $\delta > 0$ such that $\|x \| \leq \delta \Rightarrow \sup_t \left\|\varphi_t(x)\right\| \leq \epsilon$. Since $f$ vanishes at infinity there is a $K > 0$ such that $\|f(s)\| \leq \delta$ when $|s| \geq K-1$. It follows that $\left\| \alpha_0 \alpha_{j-i} f(t + \cdot - i)\right\| \leq \delta $ when $|t-i| \geq K$, and hence that 
$$
\left\|\sum_{i,j \in \mathbb Z} \varphi_i \left( \tau_i \left( \alpha_i \alpha_j f(t + \cdot)\right)\right) \otimes e_{ij} - \sum_{i,j \in \mathbb Z, |t-i| \leq K }  \varphi_{t + (i-t)} \left(  \alpha_0 \alpha_{j-i} f(t + \cdot - i)\right) \otimes e_{ij}\right\| \leq 3 \epsilon .
$$
It follows from Theorem \ref{aA} that
$$
\left\{ \alpha_0 \alpha_{j-i} f(t + \cdot - i) : \ t \in \mathbb R, i,j \in \mathbb Z\right\}
$$
is a pre-compact subset of $SA$, so we can apply v) Lemma \ref{pre} to conclude that
$$
\left\| \sum_{i,j \in \mathbb Z, |t-i| \leq K }  \varphi_{t + (i-t)} \left(  \alpha_0 \alpha_{j-i} f(t + \cdot - i)\right) \otimes e_{ij} -  \sum_{i,j \in \mathbb Z, |t-i| \leq K }  \varphi_{t} \left( \tau_t\left(  \alpha_i \alpha_{j} f(t + \cdot)\right)\right) \otimes e_{ij}\right\| \leq \epsilon
$$
for all $t$ large enough. Since $\left\| \tau_t\left(  \alpha_i \alpha_{j} f(t + \cdot)\right)\right\| = \left\|\alpha_0\alpha_{j-i} f(t + \cdot -i)\right\| \leq \delta$ when $|t-i|\geq K$, we find all together that
$$
\left\|\psi'_t(f) - \sum_{i,j \in \mathbb Z }  \varphi_{t} \left( \tau_t\left(  \alpha_i \alpha_{j} f(t + \cdot)\right)\right) \otimes e_{ij}\right\| \leq 7\epsilon
$$
for all $t $ large enough. It follows from Theorem \ref{aA} that
$$
\left\{ \tau_t(\alpha_i\alpha_j)f: \ t \in \mathbb R, i,j \in \mathbb Z\right\}
$$
is a pre-compact subset of $SA$, so we conclude from i) of Lemma \ref{pre} that 
$$
\lim_{t \to - \infty}  \sum_{i,j \in \mathbb Z }  \varphi_{t} \left( \tau_t\left(  \alpha_i \alpha_{j} f(t + \cdot)\right)\right) \otimes e_{ij} = 0 .
$$
It follows that if we set
$$
\psi''_t(f)  =  \sum_{i,j \in \mathbb Z }  \varphi_{t} \left( \tau_t\left(  \alpha_i \alpha_{j}\right) f\right) \otimes e_{ij} ,
$$
then $\psi''=\left(\psi''_t\right)_{t \in \mathbb R} : SA \to B$ is a translation invariant asymptotic homomorphism which asymptotically agrees with $\psi'$ and hence in particular is homotopic to $\psi'$. 
 


Now define a continuous path $\{\beta^{\lambda}_i\}_{i \in \mathbb Z}, \lambda \in [0,\infty[$, of partitions of unity in $C_0(\mathbb R)$ such that
$$
\beta^{\lambda}_i(\cdot) = \alpha^2_i(\cdot + \lambda), \quad i \geq 1, \quad
\beta^{\lambda}_i(\cdot) = \alpha^2_i(\cdot - \lambda), \quad i \leq -1,
$$
and
$$
\beta^{\lambda}_0(\cdot) = 1 - \sum_{i=1}^{\infty} \left( \alpha_i^2(\cdot + \lambda) + \alpha_{-i}^2(\cdot - \lambda) \right) .
$$
Set
$$
\alpha_i^{\lambda} = \sqrt{\beta_i^{\frac{\lambda}{1-\lambda}}}, \lambda \in [0,1[,
$$
and 
$$
\alpha_i^1 = 0, i \neq 0, \ \  \alpha_0^1 = 1.
$$
Note that
$$
\alpha_i^{\lambda}\alpha_j^{\lambda} = 0, \ |i -j| \geq 2,
$$
and 
\begin{equation}\label{uu13}
\sum_{ j \in \mathbb Z} \left(\alpha_j^{\lambda}\right)^2 \alpha_k^{\lambda} = \alpha_k^{\lambda}
\end{equation}
for all $\lambda \in [0,1]$ and $k \in \mathbb Z$. For $t \in \mathbb R, \lambda \in [0,1]$, set
$$
\Psi_t(f)(\lambda) =  \sum_{i,j \in \mathbb Z} \varphi_t\left( \tau_{t} \left(\alpha_i^{\lambda} \alpha_j^{\lambda}\right)f\right) \otimes e_{ij} .
$$
Since $f$ vanishes at infinity we have that
$$
\lim_{i \to \pm \infty}  \sup \left\{ \left\|\tau_{t} \left(\alpha_i^{\lambda} \alpha_j^{\lambda}\right)f\right\|: \  j \in \mathbb Z, \lambda \in [0,1] \right\} = 0,
$$
so the continuity of $\varphi_t$ at $0$ ensures that for any $\epsilon > 0$, there is a $K > 0$ such that
$$
\left\|\Psi_t(f)(\lambda) -  \sum_{i,j \in [-K,K]} \varphi_t\left( \tau_{t} \left(\alpha_i^{\lambda} \alpha_j^{\lambda}\right)f\right) \otimes e_{ij} \right\| \leq \epsilon
$$
for all $\lambda$. Since $\lambda \mapsto  \sum_{i,j \in [-K,K]} \varphi_t\left( \tau_{t} \left(\alpha_i^{\lambda} \alpha_j^{\lambda}\right)f\right) \otimes e_{ij}$ clearly is an element of $IB\otimes \mathbb K$, we conclude that $\Psi_t(f) \in IB \otimes \mathbb K$ for all $t \in \mathbb R$. Since the above approximation to $\Psi_t(f)$ can be made uniform in any compact subset of $\mathbb R$, we see that $t \mapsto \Psi_t(f)$ is continuous. We claim that $\Psi = \left(\Psi_t\right)_{t \in \mathbb R}: SA \to IB$ is asymptotic homomorphism. It follows from Theorem \ref{aA} that 
$$
\left\{ \tau_t\left(\alpha_i^{\lambda}\alpha_j^{\lambda} \right) f: \ t \in \mathbb R, i,j \in \mathbb Z, \lambda \in [0,1] \right\}
$$
is a pre-compact subset of $SA$. We can therefore conclude from i) Lemma \ref{pre} that $\lim_{t \to -\infty} \Psi_t(f) = 0$ for all $f \in SA$. If we apply iii) and iv) of Lemma \ref{aA} instead of i), and use (\ref{uu13}) we find that $\lim_{t \to \infty} \Psi_t(f)\Psi_t(g) - \Psi_t(fg) = 0$ for all $f,g \in SA$, and similar considerations regarding linearity and self-adjointness show that $\Psi$ is an asymptotic homomorphism. Similarly, an obvious application of v) of Lemma \ref{pre} shows that $\Psi$ is asymptotically translation invariant. Thus $\Psi$ gives us a homotopy of asymptotically translation invariant asymptotic homomorphisms connecting $\psi''$ to $s \circ \varphi$ and we have established the commutativity of the first triangle. To obtain it for the second, it suffices to note that $\Psi$ is translation invariant when $\varphi$ is; the algebraic condition, $\Psi_{t-s} = \Psi_t \circ \tau_s$, is trivial, and the equi-continuity follows from the same considerations we presented for $\Phi$ above.
\end{proof}


By standard arguments $s_* : \Ext_h(A,B) \to \Ext_h(A,B \otimes \mathbb K)$ and $s_* : [[SA,B]]_{a, \tau} \to [[SA, B \otimes \mathbb K]]_{a, \tau}$ are both isomorphisms (of semigroups) when $B$ is stable, so Lemma \ref{l10} and Lemma \ref{l12} yields the following:

\begin{thm}\label{th3}  Let $A$ and $B$ be $C^*$-algebras, $A$ separable, $B$ $\sigma$-unital and stable. Then $[[SA,B]]_{a, \tau} \simeq  [[SA,B]]_{\tau} $ as abelian semigroups, and
$$
CH_{\tau} : \Ext_h(A,B) \to [[SA,B]]_{ \tau} 
$$
is an isomorphism.
\end{thm}
\begin{proof} It follows from Lemma \ref{l10} and Lemma \ref{l12} that $CH_{\tau} : \Ext_h(A,B) \to [[SA,B]]_{a,\tau}$ is an isomorphism. It follows from Lemma \ref{l12} that the diagram 
\begin{equation*}
\begin{xymatrix}{
&\Ext_h(A,B\otimes\mathbb K)\ar[dr]_-{CH_\tau}&\\
[[SA,B]]_{a,\tau}\ar[ur]_-{I} \ar[drr]_(0.67){s_*}|\hole&&
[[SA,B\otimes\mathbb K]]_\tau\ar[d]\\
[[SA,B]]_\tau\ar[u] \ar[urr]_(0.3){s_*}&&[[SA,B\otimes\mathbb K]]_{a,\tau} } 
\end{xymatrix}
\end{equation*}
commutes. Since the two $s_*$-maps are both isomorphisms, we conclude that the map $[[SA,B]]_{\tau} \to [[SA,B]]_{a,\tau}$ also is. 
\end{proof}



It should be pointed out that translation invariance occurs in other forms in other (related) constructions: In \cite{H} Higson bases a proof of the Atiyah-Singer index theorem on the construction of an asymptotic homomorphism $T = \left(T^{\omega}\right)_{\omega \in [1,\infty)} : C_0(T^*M) \to \mathbb K\left(L^2(M)\right)$, where $M$ is a smooth manifold without boundary, equipped with a smooth measure, and $T^*M$ is the cotangent bundle of $M$. Since every vector space carries a canonical action $\tau$ by the multiplicative group $\mathbb R^+$, viz. $\tau_s(x) = sx$, so does $T^*M$ and in turn also $C_0(T^*M)$. It is easy to see that Higson's asymptotic homomorphism is asymptotically translation invariant in the sense that
$$
\lim_{\omega \to \infty} \left\|T^{\omega s}(a) - T^{\omega}\tau_s(a)\right\| = 0
$$
for all $a \in C_0(T^*M)$ and all $\mathbb R^+$. The same is the case of many of the asymptotic homomorphisms constructed in \cite{HKT}.


As a final remark, let us point out that $[[SA,B \otimes \mathbb K]]_{\tau}$ is known to coincide with the E-theory group $E(SA,B)$ in at least two cases: When $A$ is nuclear and when $A$ is a suspension. It remains unclear exactly how widespread this coincidence is.





\vspace{2cm}
\parbox{7cm}{V. M. Manuilov\\
Dept. of Mech. and Math.,\\
Moscow State University,\\
Moscow, 119992, Russia\\
e-mail: manuilov@mech.math.msu.su
}
\hfill
\parbox{6cm}{K. Thomsen\\
Institut for matematiske fag,\\
Ny Munkegade, 8000 Aarhus C,\\
Denmark\\
e-mail: matkt@imf.au.dk
}

\end{document}

